Author's response to reviews

Title: Patient with ring chromosome15 20-years cytogenetic and molecular follow-up: a case report.

Authors:

Roberta S Guilherme Ph.D (robertaguilherme@hotmail.com)
Vera FA Meloni Ms (vera.meloni@gmail.com)
Sylvia S Takeno Ph.D (sylvinha66@yahoo.com.br)
Renata Pellegrino Ph.D (renata.pellegrino@gmail.com)
Decio Brunoni Dr (deciobrunoni@uol.com.br)
Leslie D Kulikowski Ph.D (lesliekulik@gmail.com)
Maria I Melaragno Ph.D (melaragno.morf@epm.br)

Version: 5 Date: 7 March 2012

Author's response to reviews: see over
Dear,

Thank you for considering our MS “Patient with r(15) 20-years cytogenetic and molecular follow-up: a case report” for publication:

Enclosed please find the Revised Version of the manuscript taking into consideration we removed the figures 1a and 1b to protect the patient’s anonymity as suggested by the journal.

Hoping that now our paper meets all the requirements for publication and thanking you for all your attention.

Sincerely yours,

Maria Isabel Melaragno, Ph.D
e-mail: melaragno.morf@epm.br
Department of Morphology and Genetics
Universidade Federal de São Paulo
Rua Botucatu, 740 CEP 04023-900
São Paulo, SP, Brazil
Phone/fax: +55 11 5576-4260